Author's response to reviews

Title: Occupational skill level differences in suicide mortality: Evidence of changes over time and during the Global Financial Crisis in Australia.

Authors:

Allison J Milner (allison.milner@unimelb.edu.au)
Heather Niven (h.niven@gmail.com)
Anthony D LaMontagne (tony.lamontagne@deakin.edu.au)

Version: 2 Date: 17 March 2015

Author's response to reviews:

Carlo Rye Chua
Journal Editorial Office
BioMed Central

Dear Carlo
Thank you for your email. The changes you requested have been made.

Kind regards
Allison